Author’s response to reviews

Title: Effects of β1-adrenergic receptor blockade on the cerebral microcirculation in the normal state and during global brain ischemia/reperfusion injury in rabbits

Authors:

Nobumasa Asano (nasano@yamanashi.ac.jp)
Sohei Hishiyama (shishiyama@yamanashi.ac.jp)
Tadahiko Ishiyama (ishiyama@yamanashi.ac.jp)
Masakazu Kotoda (mkotoda@yamanashi.ac.jp)
Takashi Matsukawa (takashim@yamanashi.ac.jp)

Version: 2 Date: 11 Feb 2020

Author’s response to reviews:

February 11, 2020

PHAT-D-19-00123

Effects of β1-adrenergic receptor blockade on the cerebral microcirculation in the normal state and during global brain ischemia/reperfusion injury in rabbits

Jing Zhang

Akila Sridhar

Editor, BMC Pharmacology and Toxicology

Dear Editors:

We are pleased to submit the revised version of our manuscript to be considered for publication in your journal.

Thank you very much for your supportive and constructive comments and for giving us the opportunity to revise and resubmit this manuscript.
In response to your comments, we have revised the manuscript to minimize the textual overlap with our previous papers. The revised manuscript is the clean copy of the final version for the publication.

We hope that our manuscript will be accepted for publication in your journal and look forward to hearing from you.

Sincerely,

Masakazu Kotoda, MD PhD

Department of Anesthesiology, Faculty of Medicine, University of Yamanashi
1110 Shimokato, Chuo, Yamanashi 409-3898, Japan
Tel: +81-55-273-9690
Fax: +81-55-273-9687
Email: mktoda@yamanashi.ac.jp

Response to the Editor Comments:

1. Textual overlap

We note that the current submission contains some textual overlap with other previously published works, in particular:

“https://bmcanesthesiol.biomedcentral.com/articles/10.1186/s12871-017-0331-5”

And


This overlap mainly exists in the Methods.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.

> We have revised the manuscript to minimize the textual overlap with our previous papers.

2. Clean copy

Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean
copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

>The uploaded revised manuscript is the clean copy of the final version for the publication. We have removed all highlights and track changes from the revised manuscript.